Author's response to reviews

Title: Influence of parental employment status on Dutch and Slovak adolescents' health

Authors:

Maria Sleskova (maria.sleskova@upjs.sk)
Jolanda Tuinstra (j.tuinstra@drenthe.nl)
Andrea Madarasova Geckova (geckova@upjs.sk)
Jitse P van Dijk (j.p.van.dijk@med.umcg.nl)
Ferdinand Salonna (nando@upjs.sk)
Johan W Groothoff (j.w.groothoff@med.umcg.nl)
Sijmen A Reijneveld (s.a.reijneveld@med.umcg.nl)

Version: 3 Date: 5 October 2006

Author's response to reviews:

Title: Influence of parental employment status on Dutch and Slovak adolescents' health
Authors: Maria Sleskova, Jolanda Tuinstra, Andrea Madarasova Geckova, Jitse P van Dijk, Ferdinand Salonna, Johan W Groothoff, Sijmen A Reijneveld

Dear Editors,

We have the pleasure to send you the second revision of our manuscript. We have made several changes according to the requirements of editor and some according to comments of Reviewer 2.

The changes are as follows:
Editor:
- Manuscript has been checked to be in line with "manuscript formatting checklist"
- Abstract has been structured into Background, Methods, Results and Conclusions
- Details about the procedure of data collection and ethical approval of the study have been added.
- Placement of tables in the text have been removed.
- References have been formatted using Reference Manager.
Reviewer 2:
- The text about dichotomisation of self-rated health and psychological well-being has been extended to make it clearer for readers.
- Table titles have been changed where it was necessary.
- Conclusions both in abstract and in the manuscript have been changed. Clear statements about what was found have been added.

With kindest regards
On behalf of the co-authors

Maria Sleskova, MSc.
Institute of Philological and Social Sciences
PJ Safarik University
Moyzesova 16
040 01 Kosice
Slovak Republic
+421 55 2342127
maria.sleskova@upjs.sk